Author's response to reviews

Title: Pay for Performance: Will Dentistry Follow?

Authors:

Andreea Voinea-Griffin (avoineagriffin@yahoo.com)
Jeffrey L Fellows (jeffrey.fellows@kpchr.org)
D. Brad Rindal (d.brad.rindal@healthpartners.com)
Andrei Barasch (abarasch@uab.edu)
Gregg H Gilbert (ghg@uab.edu)
Monika M Safford (msafford@uab.edu)

Version: 2 Date: 8 September 2009

Author's response to reviews: see over
Melissa Norton, MD  
Editor-in-Chief  
BMC Oral Health

September 7, 2009

Dear Dr. Norton,

The attached manuscript explores the issue of Pay for Performance (P4P) in dentistry. Since professional acceptance will be enhanced if payment methodology is linked to quality, we approached P4P implementation from a quality of care perspective. Our intent is to provide a fresh look to what the profession needs to accomplish to build the prerequisites for quality, performance and possibly P4P. Kindly consider this manuscript for publication in BMC Oral Health.

If you have any questions, please do not hesitate to contact me by email, at avoineagriffin@yahoo.com or phone: 205-532-8890.

Sincerely,

Andreea Voinea-Griffin, DDS, MBA, MSHA  
Corresponding author